Author’s response to reviews

Title: Classification of pediatric functional gastrointestinal disorders related to abdominal pain using Rome III vs. Rome IV criterions

Authors:
Trent Edwards (trent.edwards@ttuhsc.edu)
Craig Friesen (cfriesen@cmh.edu)
Jennifer Schurman (jschurman@cmh.edu)

Version: 1 Date: 26 Oct 2017

Author’s response to reviews:

The following changes were made:

1. Key words were added to the end of the abstract (pg. 3).

2. A STROBE checklist was completed.

3. A statement was added regarding a consent waiver in the Ethics and Consent to Participate section (pg. 14).

4. The reason for the data not yet being publicly available was added to the Availability of Data and Materials section (pg. 15).

5. An Acknowledgements section was added (pg. 15).